Contents lists available at ScienceDirect



Reliability Engineering and System Safety

journal homepage: www.elsevier.com/locate/ress

A value of prediction model to estimate optimal response time to threats for accident prevention





Tiantian Zhu^{a,*}, Stein Haugen^a, Yiliu Liu^b, Xue Yang^c

^a Department of Marine Technology, Norwegian University of Science and Technology, Trondheim, Norway

^b Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, Trondheim, Norway

^c Navigation College, Dalian Maritime University, Dalian, China

ARTICLE INFO

Keywords: Response time Decision time Accident prediction Prediction horizon Value of prediction Value of information Accident prevention

ABSTRACT

This paper presents a novel value of (imperfect) prediction (VoP) model to estimate optimal response time to a threat that may result in an accident. The proposed VoP model is based on information value theory and considers both prediction accuracy and action failure probability over time. The optimal response time is dependent on parameters: the ratio between the accident cost and response action cost, accident probability, action failure probability, prediction performance, and response strategy (a series of sequential responses or a single response). A case study of iceberg management is presented to demonstrate the proposed approach; a sensitivity study is done to evaluate how optimal response time changes with those parameters. The case study show that it is reasonable to respond as early as possible if the threat can lead to a serious accident, while the response can be postponed when the potential consequence is moderate. In addition, the proposed VoP model is proven able to calculate accuracy requirements, thresholds for tolerating risk and acting precautionarily, and maximum investment in accident prevention. Imperfect prediction can lower risk acceptance threshold and higher the threshold of being precautionary; and it is reasonable to increase action cost.

1. Introduction

1.1. Background

When an iceberg is approaching an offshore installation, it is not clear whether a collision will happen. For such a threat that has a potential to damage an asset of interest, the decision-maker needs to decide how to respond to avoid accidents with the lowest cost. For example, one can tow the iceberg away [1] immediately after the iceberg is observed, to avoid any risks, but it is unnecessary in most cases since the chance of collision is very low. Thus, after receiving the first signals of a threat (e. g., seeing an iceberg), the decision-maker does not need to respond until she/he has higher confidence that the threat will make a damage. Situation changes rapidly, and validity of prediction is also time-varying [2, 3]. When prediction is made early, the signal for anticipating either an accident or no accident may be very weak [4], or there is too much noise and too many possible future scenarios. The prediction accuracy may be insufficient to support decision-making. Short-term prediction tends to be more accurate, and it may be worthwhile to postpone the response. However, relying on short-term prediction leaves little time for action implementation; a failed action may lead to a disastrous result.

Therefore, after observing a threat, a question is raised: when is the optimal time to respond? In other words, what is the optimal prediction horizon to use? We call this as the "optimal responding problem". It is a question to both decision-makers and the academics because a responding decision must be made under partial predictability where it is possible to take observations and predict the behavior of the threat (e. g., the iceberg) but only to a certain degree. Here, the response time t_r is a time interval and it is defined in relation to time t_t when the hazardous situation terminates as illustrated by Fig. 1. The time point of response t_{pr} is the time to make & accept a prediction and implement a given response action to prevent accident. By default, a response includes making a prediction based on the information available, decisionmaking based on the prediction result, and a potential following action to reduce risks. An actual action will follow the decision-making if an accident is predicted to occur while no actual action is implemented if no accident is predicted.

Determining the optimal time to respond is a generic problem and is valuable for the planning of time-critical and expensive response

 * Corresponding author at: Jonsvannsveien 82, NTNU, Trondheim 7050, Norway.

E-mail addresses: tiantian.zhu@ntnu.no (T. Zhu), yiliu.liu@ntnu.no (Y. Liu), xue.yang@dlmu.edu.cn (X. Yang).

https://doi.org/10.1016/j.ress.2022.109044

Received 23 August 2021; Received in revised form 11 December 2022; Accepted 13 December 2022 Available online 14 December 2022 0951-8320/© 2022 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/). actions, risk prediction and control during operation (e.g., flood [5], ship collision [6, 7], vehicle collision [8]), design of predictive and intelligent alarms [9], threat management, emergency response [10–12] (such as when we stop watching a hurricane and start evacuation), system health monitoring and response [13–15]. Ambiguity in when to respond can lead to action delay and eventually cause incident or accident [16]. Solving this problem also bring benefits to action planning in autonomous systems which can take in and process information from the environment to make decisions themselves.

In practices, the problem of when to respond to a threat is resolved by risk monitoring, experience, or regulation rules. An example is response to ships on collision course with an offshore installation in the North Sea, where there are fixed time limits for various response actions (alarm, mustering, evacuation). As for how rules are set, the rationale is often not very clear. When it comes to risk monitoring, a risk threshold is usually assumed to exist to trigger an response action [17]. Such a threshold is like an alarm trip point. If risk exceeds the threshold, action should be taken, otherwise not. To formulate a good rationale to determine when to respond to threat, how far in the future we are predicting/monitoring the risk, how soon the threat will propagate into an accident (which is called the process safety time in the process industries [18]), or the availability of actions when the threshold is breached [19] should also have been considered. Overall, rationale for when to respond and approach to derive optimal response time are needed so that risk can be further minimized, and system safety can be maintained at a better cost-efficient manner.

1.2. Time of response

In the field of psychology and neuroscience, speed-accuracy tradeoff [20, 21] is used to describe the complex relationship between an individual's willingness to respond slowly and make relatively fewer errors compared to the willingness to respond quickly and make relatively more errors. The time interval until the decision-maker decide from the presentation of stimulus is called the choice reaction time [22–24]. From the information theory and behavioral decision-making points of view, there is a cost in spending time for information accumulation [25, 26]. A decision-maker tends to respond later if accuracy is emphasized and tends to decide early if speed is emphasized. If high discriminability can be reached early, the decision-maker is able to decide early [27].

Following the speed-accuracy tradeoff, this paper proposes a novel rationale, and a mathematical problem formulation to determine when to respond to threat. The key to determine whether to respond immediately or to postpone is whether future updated information will give a more reliable prediction to warrant the postponing; what should be managed is the tradeoff between decreased cost by increased prediction accuracy, and increased cost by acting late, as illustrated in Fig. 2. The optimal time to respond is when the marginal expected gain of prediction is equivalent or smaller than the marginal cost of postponing action.

Information value theory has been used to quantify the added value from information and support rational decision making under uncertainty [28–34]. Value of information (VoI) is recommended for sensitivity analysis for reliability analysis [33], probabilistic safety assessment [29] and safety management [35]; here, a sensitivity analysis is to evaluate which input variables one should collect more information to reduce their uncertainty. VoI is also used for making testing/inspection and maintenance decision [14, 32]. Prediction is to provide information. The value of prediction can be evaluated in a similar manner of information. A Value of (imperfect) Prediction (VoP) method based on information value theory is formulated to calculate the optimal response time quantitatively. The gain from an improved prediction can be evaluated by the cost reduction when the predicted result is used in decision-making. The cost of acting later can be modelled by the increased probability of failure of the action due to reduced time available. The optimal time to respond can be found by minimizing the total cost as the combination of gain from improved prediction by postponing and loss from acting later. This paper does the following:

- Develops a utility model of the tradeoff phenomenon.
- Proposes a solution for the "optimal responding problem" by modeling the tradeoff.
- Formulates the "optimal responding problem" and its solution through probabilistic mathematics. Such formulation provides a unified manner to describe and solve the problem and enables a general application of the solution.

1.3. Potential contributions of this research

This research provides a novel rationale to determine when to respond provided in this paper can reduce ambiguity in the timing of decision-making, thus, lower the possibility of decision error [16]. The methodological novel contribution of this research can be expected from the formulation of the "optimal responding problem" and the VoP model as a solution. Thus, the optimal response time can be derived.

In addition, the method proposed in this paper can also be used to answer the following questions:

- Whether can we rely on our prediction?
- How good must the prediction be to be usable or to provide additional value?
- How much can we invest in the response action?
- What is the risk threshold between making decisions based on prediction and making decisions without prediction (such as taking a precautionary approach or simply tolerating the risk)?

In the remainders of this paper, Section 2 describes the problem and its assumptions, mathematical representation, and proposed solution. Section 3 presents additional theoretical applications of the proposed VoP model. Section 4 describes a case study and result obtained by using the proposed method. Section 5 includes implications and discussions; and finally, Section 6 concludes the study.



Fig. 1. Response time (time is not represented to scale).



Fig. 2. Illustration of trade-off between prediction accuracy and action failure probability over time.

2. Mathematical description of the problem and solution

2.1. The optimal responding problem

To find the optimal response time, we can have the following general assumptions:

- 1. The threatening event terminates at time *t*_b either because an accident occurs, or the threat is not relevant anymore.
- 2. The system state in the future is unknown. It is not clear whether an accident will happen or not until the last moment. However, the (statistical) probability of occurrence can be described.
- 3. The system state in the future is partially predictable, which means that the prediction of future system state is imperfect, and accuracy will be not 100%.
- 4. Predictability of system state improves when prediction horizon (how far ahead we predict) is shorter. Namely, we are more confident to discriminate "no accident" and "accident" when the threat is approaching.
- 5. Costs and chances of failure (fail to eliminate the threat under the period of time t_r available) of potential actions are known.
- 6. Under a certain prediction horizon t_r , the prediction model predicts whether the accident will happen or not (the prediction output is binary: "yes" or "no") at time t_r in the future. For instance, if the prediction horizon is 3 h, the model predicts whether the accident will occur or not in 3 h.
- 7. The decision-maker will respond immediately according to the prediction if the decision-maker trusts the prediction, meaning no time gap between prediction and response action.
- 8. The decision-maker only uses the result of one prediction to determine whether to implement the response action or not. Since predictions always are made based on accumulated and updated information, earlier predictions can never be more reliable than the later one. Using multiple (redundant) predictions will not improve accuracy either.
- 9. It is not possible to change the response after a decision has been made.

The termination time t_t may be known or not known. In the cases when we are only interested in a specific time, for example a deadline, or the execution time of a planned activity, the termination time t_t is considered known. For continuous operation, we are concerned with the whole operation period. We do not know when an accident will occur, and the termination time t_t is considered unknown.

- For the first scenario when the termination time t_t is known, the solution could tell us the optimal time is to respond (*t* before the specific termination time point t_t).
- For the second scenario when the termination time *t_t* is not known, a monitoring process is introduced assuming that the termination time is *t_r* time period ahead of current time in the future. The response

action will only be taken when the prediction predicts "accident". The solution tells us what the best prediction horizon is for the monitoring.

When the prediction horizon ${\bf T}$ is a list of discrete values, the following elements are considered:

- **S** is a set of states, preliminarily expressed by ["accident", "no accident"], corresponding to accidental scenario and safe scenario. The cost associated with each state is expressed by *C*_{acc} and *C*_{no acc} respectively.
- **R** is a set of response action alternatives $[r_1, r_2, ..., r_n]$, and their costs of implementation.
- **Pr**_i is a set of failure probabilities of action alternative *r*_i under various periods of time available.
- P is a set of prediction performances for different prediction horizons. True positive rate (TPR) and false positive rate (FPR) are used as prediction performance measurements. TPR is known as sensitivity, recall or probability of detection, while FPR is known as probability of false alarm. TPR and FPR can be derived from the prediction error matrix (also called confusion matrix) [36].
- T is a set of time (prediction horizon) alternatives.
- *p_{acc}* is the occurrence probability of an accident at termination time *t_t* conditional on threat presence; correspondingly, 1-*p_{acc}* is the occurrence probability of no accident.

The system starts in a safe state but with a probability p_{acc} to end up in an accidental scenario at termination time t_t . The decision-maker receives a result from the prediction model with a prediction horizon $t_r \in \mathbf{T}$, which tells whether the accident would occur or not at the future time point t_t (t_r time period into the future after prediction is made). If the prediction model predicts "accident", the decision-maker needs to take an action $r \in \mathbf{R}$ to prevent the system moving into the accident state. The failure probability of the chosen action is $p_{t,r}$. If the prediction model predicts "no accident", the decision-maker will not do anything. As shown in Fig. 2, the predicting and responding time will affect the cost, and the decision-maker needs to find an optimal response time t_r before the termination time t_t to minimize the expected cost of such threatening scenario. If there are multiple response actions available, the final solution of this optimal responding problem is a pair of t_r and rsince the response time t_r is dependent on the response action r also. If t_r variable is continuous instead of discrete in the problem formulation, then p_r can be expressed as a function of t_r for response action r. TPR and FPR also can be expressed as functions of t_r . The notations used for the mathematical formulation are presented in Table 1.

2.2. Value of prediction (VoP) model

Information has properties like degree of precision, quality, and utility [37]. The same applies for prediction models (not limited to computational models). The Value of Information (VoI) [28] is usually

Table 1 Notations

VoP

 C_{acc}

Dace

t,

p,

 p_{t_r}

 C_r

TPR

 $C_{no acc}$

Notations

response is early enough for the action failure probability to be 0.

$$C(\text{ignore the risk}) = (1 - p_{acc}) * C_{no \ acc} + p_{acc} * C_{acc}$$

$$= p_{acc} * C_{acc}$$
(1)

C(be precautionary $|t_r, r) = (1 - p_{acc}) * C_{no acc} + p_{acc} * p_{t_r,r} * C_{acc} + C_r$

$$= p_{acc} * p_{t_r,r} * C_{acc} + C_r \tag{2}$$

The optimal response should be the option with the minimum cost among C(be precautionary| t_r , r), C(ignore the risk), and $\min(C(\text{with prediction}|t_r, r))$. The objective of the optimal responding problem is to find the t_r for a given action r which minimize expected cost as shown by Eq. (3).

Optimal
$$C(t_r, r) = \min \begin{cases} C(be \ precautionary|t_r, r) \\ C(\text{ignore the risk}) \\ \min(C(\text{with prediction}|t_r, r)) \end{cases}$$
 (3)

For a two-states ("accident, "no accident") system, there are four kinds of scenarios considering the prediction model is imperfect. The four kinds of scenarios are:

- True Positive (TP): actual "accident" state which is correctly predicted as "accident".
- False Negative (FN): actual "accident" state which is wrongly predicted as "no accident".
- False Positive (FP): actual "no accident" state which is wrongly predicted as "accident".
- Ture Negative (TN): actual "no accident" state which is correctly predicted as "no accident".

The cost of each scenario is not same. Fig. 4 illustrates that the decision process and all the possible consequences after a response decision is made based on a prediction with prediction horizon t_r . Both the distribution of the four prediction scenarios and the distribution between successful & unsuccessful action are dependent on t_r . The expected cost C(with prediction) can be expressed conditionally on taking the response action r at t_r time ahead of t_r as shown by Eq. (4).

$$C(\text{with prediction}|t_{r}, r) = Expected\left(C_{t_{r}}^{TN} + C_{t_{r}}^{FN} + C_{t_{r,r}}^{FP} + C_{t_{r,r}}^{TP,S} + C_{t_{r,r}}^{TP,F}\right)$$
(4)

In Eq. (4), $C_{t_r}^{TN}$ is the cost of true negative (TN) prediction with cost $C_{no\ acc}$; $C_{no\ acc}$ can assumed to be 0 because there is no accident; $C_{t_r,r}^{FP}$ represents the cost of false positive (FP) prediction taken with a response time t_r ; $C_{t_r}^{FN}$ is the cost of false negative (FN) prediction; $C_{t_r,r}^{TP,S}$ represents the cost of true positive (TP) prediction and action r is successful while $C_{t_r,r}^{TP,F}$ represents the cost of true positive (TP) prediction but the action r fails. To calculate the expected cost, the percentages of the four prediction results can be obtained from the confusion matrix [36]. Each cell in the confusion matrix shows the number of false positives, false negatives, true positives, and true negatives like in Table 2. Prediction performance measurements true positive rate (TPR, see Eq. (5)), false



Fig. 3. Illustration of decision tree for determining whether to apply prediction and which prediction horizon to use.

4

 TPR_{tr}
 True positive rate of the predictive model with prediction horizon tr.

 FPR
 False positive rate.

 FPR_{tr}
 False positive rate of the predictive model with prediction

Descriptions

this paper.

Response action

True positive rate.

Value of (imperfect) prediction

Occurrence probability of the accident.

time available for action implementation.

Failure probability of the response action r.

interval t- is available for implementation.

response action A, B respectively.

of accidental scenario.

Cost of accident which is used to describe the consequence

Cost when there is no accident. Cno acc is assumed to be 0 in

Response time; it is equivalent to prediction horizon and

Failure probability of response action r when there is time

The cost of response action r. CA, CB represent cost of

 $LR+ Positive likelihood ratio of the predictive model with prediction in the predictive model. It is the ratio between TPR and FPR. C(with prediction|t_r, r) Expected cost of the threat response activity if response action r is the candidate response action and prediction made with prediction horizon t_r. C(be precautionary|t_r, Expected cost of being precautionary and respond with r) action r at time t_{pr} with t_r time period available for action implementation. C(ignore the risk) Expected cost if the risk is neglected.$

calculated by the following formula:

VoI = "the value of decision situation with additional information" - "the value of current decision situation"

In this paper, since the information comes from prediction and the value is provided by cost reduction, we propose the concept of VoP based on VoI as:

VoP = "the cost of current decision situation without prediction" - "the cost of decision situation with prediction"

The optimal time to respond arrives when VoP does not increase anymore, in another word, when the marginal VoP is 0 if the marginal VoP is a monotonic function with time, where:

Marginal *VoP* (Δ *VoP*) = "the cost of current decision situation with prediction" - "the cost of decision situation with new prediction with a unit time shorter prediction horizon"

When prediction model is not used, the decision maker has the option of being precautionary and taking action immediately or the option of tolerating the risk so that no action is taken, as illustrated in Fig. 3. Application of a precautionary approach [38, 39] may lead to waste of resources. When the risk is too low for a response action, ignoring the risk is preferred. The cost of ignoring the risk is expressed by Eq. (1). The cost of being precautionary is expressed by Eq. (2). Failure probability of the response action can be assumed to be 0 when the response action is taken early. Note that respond precautionarily does not mean that the



Fig. 4. Decision tree diagram involving prediction for threat management.

Table 2 Confusion matrix.

		Actual	
		Accident (positive)	No accident (negative)
Predicted	Accident (positive) No accident (negative) Sum	# True Positive (TP) # False Negative (FN) # Positive (P)	# False Positive (FP) # True Negative (TN) # Negative (N)

positive rate (FPR, see Eq. (6)) and prediction accuracy (ACC, see Eq. (8)) can be derived from the confusion matrix. Positive likelihood ratio (LR+) (see Eq. (7)) which is used to assess the usefulness of a prediction [40] can also be derived.

True positive rate:

$$TPR = \frac{TP}{TP + FN} = \frac{TP}{P}$$
(5)

False positive rate:

$$FPR = \frac{FP}{FP + TN} = \frac{FP}{N}$$
(6)

Positive likelihood ratio (LR+):

$$LR + = \frac{TPR}{FPR}$$
(7)

Accuracy:

$$ACC = \frac{TN + TP}{FN + TN + TP + FP} = \frac{TN + TP}{P + N} = p_{acc} * TPR + (1 - p_{acc}) * (1 - FPR)$$

where $p_{acc} = \frac{P}{P + N}$; it is also called prior accident probability

(8)

To include prediction performance measurement, the cost function can be altered into:

$$C \text{ (with prediction)} = Expected \left(C^{TN} + C^{FP} + C^{FN} + C^{TP,S} + C^{TP,F}\right)$$
$$= \frac{TN}{P+N} * C_{no\ acc} + \frac{FN}{P+N} * C_{acc} + \frac{FP}{P+N} * C_r$$
$$+ \frac{TP}{P+N} * \left[(1-p_r) * C_r + p_r * (C_r + C_{acc})\right]$$

$$= \frac{1}{P+N} \{TN * C_{no \ acc} + FN * C_{acc} + FP * C_r + TP \\ * [C_r + p_r * C_{acc}] \}$$

$$= (1 - p_{acc})$$

$$* \left[\frac{TN}{TN + FP} * C_{no\ acc} + \frac{FP}{TN + FP} * C_r \right] + p_{acc}$$

$$* \left\{ \frac{FN}{TP + FN} * C_{acc} + \frac{TP}{TP + FN} * [C_r + p_r * C_{acc}] \right\}$$

$$= (1 - p_{acc}) * [(1 - FPR) * C_{no \ acc} + FPR * C_r] + p_{acc} \\ * \{(1 - TPR) * C_{acc} + TPR * [C_r + p_r * C_{acc}] \}$$

$$= (1 - p_{acc}) * [C_{no \ acc} + FPR * (C_r - C_{no \ acc})] + p_{acc} * [C_{acc} + TPR * (C_r + (p_r - 1) * C_{acc})]$$

$$= (1 - p_{acc}) * C_{no \ acc} + p_{acc} * C_{acc} + (1 - p_{acc}) * FPR * (C_r - C_{no \ acc}) + p_{acc} * TPR * (C_r + (p_r - 1) * C_{acc})$$
(9)

 $C_{no\ acc}$ is assumed to be 0. The cost function in Eq. (9) is a linear function with 6 parameters: p_{acc} , C_{acc} , *FPR*, *TPR*, p_r , C_r ; p_{acc} and C_{acc} are the probability and cost of accident respectively. p_r and C_r are the failure probability and cost of response action r. The derivative of the expected cost of each input parameter is constant but associated with other input parameters meaning that the importance of each parameter to the expected cost is dependent on other parameters. For example,

$$\Delta C(\text{with prediction})/\Delta FPR$$

$$= (1 - p_{acc}) * (C_r - C_{no \ acc}). \text{ The importance of FPR to the expected cost}$$

$$= (1 - p_{acc}) * C_r$$

C (with prediction) is dependent on the value of p_{acc} and C_r

Since *FPR* and *TPR* are dependent on the response time t_r , and p_r is dependent on the response time t_r and the chosen response action, the cost function *C* can be altered into *C*(with prediction $|t,r\rangle$ in Eq. (10).

$$C(\text{with prediction}|t, r) = p_{acc} * C_{acc} + (1 - p_{acc}) * FPR_{t_r} * C_r + p_{acc} * TPR_{t_r} \\ * (C_r - (1 - p_{t_r, r}) * C_{acc})$$
(10)

When only t_r and its corresponding FPR_{t_r} , TPR_{t_r} , and $p_{t_r,r}$ are variants, minimizing C(with prediction $|t_r, r)$ is equivalent to minimize:

$$FPR_{t_r} + TPR_{t_r} * \frac{p_{acc}}{1 - p_{acc}} * \left(1 - \left(1 - p_{t_r,r}\right) * \frac{C_{acc}}{C_r}\right)$$

 FPR_{t_r} and TPR_{t_r} are conditional on the prediction horizon and model used for prediction. They can be expressed as a function of time and

prediction model, as shown by Eq. (11) and (12) correspondingly. According to the overall assumption, the prediction performance improves with shorter prediction horizon. The prediction improvement can be a result of multisensory integration and information accumulation cross over time. Therefore, FPR_{t_r} should decrease, while TPR_{t_r} should increase with shorter prediction horizon t_r .

$$FPR_{t_r} = f(t_r, model)$$
, with constraints $0 \le FPR_{t_r} \le 1$. (11)

$$TPR_{t_r} = g(t_r, model)$$
, with constraints $0 \le TPR_{t_r} \le 1$. (12)

The failure probability of a response action is dependent on the time available for implementation. Shorter time available implies higher failure probability. The relation between time and probability may be linear, but more complex relationships are also possible. A logarithmic function and cumulative gamma distribution allow flexibility to model different relations by varying their parameters. In Eq. (13), a logarithmic function is used to express the relation between failure probability and time available. The Gamma distribution automatically constrains the failure probability between [0,1]. Eq. (14) express the action failure probability using the Gamma distribution.

$$p_{t_r,r} = \{ b - d \log_x(t_r - a + 1), if a \le t_r < c \\ 1, if t_r < a \end{cases}$$
(13)

or

$$p_{t_r,r} = 1 - \frac{1}{\Gamma(\mathbf{k})} \gamma \left(k, \ \frac{(t_r - a)}{\theta} \right)$$
(14)

The failure probability function and its parameters can be estimated from historical data, experimental data or from a physical understanding of the phenomenon. The failure probability of a response action can be dependent on factors such as equipment and manpower availability and threat condition. Conditional failure probability functions on other factors can be derived for real application.

When prediction is perfect (TPR = 1, FPR = 0), there will be no wasted actions (false positive prediction) and no undetected accident cost (false negative prediction). The only risk is from failure of response

2.3. Cost function for a series of responses (a series of predictions and actions)

An underlying assumption in the previous section is that there is only one response, namely only one prediction will be made, and one action will be taken. Such an assumption is often challenged. In many cases, the second action may be possible to be taken if the first action fails, formulated as strategy 1 below. An accident will occur only if both the first action and the remedial action fail. In addition, we can also make a second prediction if the first action fails and proceed further based on the second prediction with a shorter prediction horizon (higher accuracy because of new information available) to reduce unnecessary response action in case the first prediction is false positive (predicted "accident" but actual "no accident"), which is formulated as the strategy 2. New information can be e.g., that we see that the first action does not have the effect that we expected, other characteristics of the situation change in unanticipated ways. Before an accident occurs, further information may tell the decision-maker that the first prediction was false negative (predicted "no accident" but actual "accident") and allow another response. For severe accidents with high consequence, the cost of false negative is very high. To reduce the possibility of false negative, we can take a second prediction to double check and make decision based on the second prediction, which is formulated as strategy 3. The time interval between the first and the second prediction should be long enough that possible new information will be available.

The response can be extended to a series of responses, as shown in Fig. 5. Five new cost functions are obtained as shown below. In this paper, we only consider a series of two actions, no further action C, action D etc. can be initiated after action B fails. The principle will be the same if more response actions are considered thus not discussed in this paper. To take action B, we assume that we can observe action A fails after some time so that it is possible to start action B. When it comes to prediction, the second prediction can be made after some time if the first prediction predicts "no accident" or can be made when action A fails or in either situation.

Strategy 1: Take action B without taking a second prediction after action A fails.

$$C(\text{strategy 1}) = (1 - p_{acc}) * FPR_{t_{r1}} * (C_A + p_{t_{r1},a} * C_B) + p_{acc} * \{(1 - TPR_{t_{r1}}) * C_{acc} + TPR_{t_{r1}} * [C_A + p_{t_{r1},a} * C_B + C_{acc} * p_{t_{r1},a} * p_{t_{r2},b}]\}$$

action. The potential value of prediction meaning cost reduction is maximized. The Value to Invest (VtI) in developing and improving the prediction model can be determined by the potential cost reduction through perfect prediction and number of applications. The expected

Strategy 2: Take action based on the second prediction after action A fails.

$$C(\text{strategy 3}) = (1 - p_{acc}) * \left[FPR_{t_{r1}} * (C_A + p_{t_{r1},a} * FPR_{t_{r2}} * C_B) \right] \\ + p_{acc} * \left\{ (1 - TPR_{t_{r1}}) * C_{acc} + TPR_{t_{r1}} * \left[C_A + p_{t_{r1},a} * \left((1 - TPR_{t_{r2}}) * C_{acc} + TPR_{t_{r2}} * \left(C_B + p_{t_{r2},b} * C_{acc} \right) \right) \right] \right\}$$

cost of perfect prediction is expressed in Eq. (15).

$$C(\text{perfect prediction}|t_r, r) = p_{acc} * (C_r + C_{acc} * p_{t_r, r})$$
(15)

Strategy 3: Predict for the second time if the first prediction predicts "no accident" and respond based on the second prediction.



$$C(\text{strategy 2}) = (1 - p_{acc}) * [(1 - FPR_{t_{r_1}}) * FPR_{t_{r_2}} * C_B + FPR_{t_{r_1}} * C_A] + p_{acc} * \{(1 - TPR_{t_{r_1}}) * [(1 - TPR_{t_{r_2}}) * C_{acc} + TPR_{t_{r_2}} * (C_B + p_{t_{r_2,b}} * C_{acc})] + TPR_{t_{r_1}} * [C_A + p_{t_{r_1,a}} * C_{acc}]\}$$

Strategy 4: Predict for the second time if action A fails or the first prediction predicts "no accident".

is made with a prediction horizon of t_{r3} if the first prediction predicts "no accident". Consequently, t_{r2} and t_{r3} must be smaller than t_{r1} . With strategy 5, the first response is taken precautionarily without prediction.

$$C(\text{strategy 4}) = (1 - p_{acc}) * [(1 - FPR_{t_{r1}}) * FPR_{t_{r3}} * C_{c} + FPR_{t_{r1}} * (C_{A} + p_{t_{r1},a} * FPR_{t_{r2}} * C_{B})] + p_{acc} * \{(1 - TPR_{t_{r1}}) * [(1 - TPR_{t_{r3}}) * C_{acc} + TPR_{t_{r3}} * (C_{A/B} + p_{t_{r3},a/b} * C_{acc}) + TPR_{t_{r2}} * [C_{A} + p_{t_{r2},a} * ((1 - TPR_{t_{r2}}) * C_{acc} + TPR_{t_{r3}} * (C_{B} + p_{t_{r3},b} * C_{acc}))]\} \}$$

Strategy 5: Be precautionary for the first response action and use prediction later if the first action fails.

Therefore, there is no $TPR_{t_{r1}}$, $FPR_{t_{r1}}$ in the cost function. When considering a series of responses, the optimal time for the earlier response is dependent on the planned later responses except when a precautionary

$$C(\text{strategy 5}) = C_A + p_{t_{r_1,a}} \\ * \{ (1 - p_{acc}) * [(1 - FPR_{t_{r_2}}) * FPR_{t_{r_3}} * C_B + FPR_{t_{r_2}} * C_B] + p_{acc} \\ * \{ (1 - TPR_{t_{r_2}}) * [(1 - TPR_{t_{r_3}}) * C_{acc} + TPR_{t_{r_3}} * (C_B + p_{t_{r_3,b}} * C_{acc})] + TPR_{t_{r_2}} * [C_B + p_{t_{r_2,b}} * C_{acc}] \} \}$$

With action B available, the cost of being precautionary is also different. The new cost function is:

C(being precautionary | two actions)

$$= (1 - p_{acc}) * C_{no\ acc} + p_{acc} * p_{t_{r1},\ a} * p_{t_{r2},b} * C_{acc} + p_{t_{r1},a} * C_B + C_A$$
(16)

In the extended cost functions, we use:

- C_A is the cost of first response action A.
- $p_{t_{r1},a}$ is the failure probability of action A with t_{r1} time period available.
- C_B is the cost of second response action B if the response action fails.
- $p_{t_{r_2},b}$ is the failure probability of action B with t_{r_2} time period available.
- $C_{A/B}$ is the cost of either response action A or B.
- $p_{t_{r_3,d/b}}$ is the failure probability of action A or action B with t_{r_3} time period available.
- *TPR*_{*t*_{c1}}, *FPR*_{*t*_{c1}}, *TPR*_{*t*_{c2}}, *FPR*_{*t*_{c3}}, *FPR*_{*t*_{c3}} are the prediction performance with prediction horizon *t*_{*r*1}, *t*_{*r*2}, *t*_{*r*3}.

With strategy 4, the first prediction is the originally assumed one; the second prediction with prediction horizon t_{r2} is made if action A fails or

approach is taken. An optimization of the later prediction and action is needed to obtain an overall optimal response. To calculate the best time to decide, a multivariate optimization algorithm, greedy searching method, or dynamic programming is needed.

3. Additional applications of the VoP model

3.1. Whether we should rely on prediction

Even though we know that the prediction is imperfect, we may still use it because it provides additional value or is better than not using it. Therefore, the key question is whether the prediction can improve our decisions. From Eq. (17), it can be found that to make the prediction provide additional value compared with simply ignoring the risk or being precautionary, the VoP should be at least greater than 0 or greater than the cost of applying the prediction model. If VoP > 0, then we should rely on the prediction, otherwise, not.

VoP = min(C(ignore the risk), C(be precuationary)) - C(with prediction)(17)

Positive likelihood ratio LR+ is one measurement of the usefulness of prediction, as calculated by Eq. (7) in Section 2.2. However, positive likelihood ratio LR+ does not consider the consequences associated with



Fig. 6. Feasible regions of TPR & FPR which makes VoP >0.

different prediction results or the possibility of not relying on prediction. We therefore suggest prediction effectiveness as formulated by Eq. (18) to measure the usefulness of a prediction model. If the prediction effectiveness is larger than 1, the result from prediction is useful. Prediction effectiveness (PE) potentially can be used to evaluate the quality of diagnostic techniques.

$$PE = \frac{\min(C(\text{ignore the risk}), C(\text{be precautionary}))}{C(\text{with prediction})}$$
(18)

The constraint function VoP > 0 can be also used to calculate prediction performance requirement (Section 3.2), the risk thresholds (Section 3.3), and the maximum investment in a response action (how much we can spend to prevent an accident with a known consequence) (Section 3.4).

3.2. Prediction performance requirement

At any time for a known threat and given response action, the minimal prediction performance requirement is that the FPR & TPR pair makes VoP > 0. The FPR &TPR requirement is dependent on three factors: accident probability, action failure probability, and the ratio between accident cost and action cost, as shown in Eq. (19).

Constraint function:

VoP > 0

yields

$$\begin{aligned} \min(p_{acc} * C_{acc}, \ p_{acc} * p_r * C_{acc} + C_r) \\ &- [p_{acc} * C_{acc} + (1 - p_{acc}) * FPR * C_r + p_{acc} * TPR * (C_r - (1 - p_r) * C_{acc})] \rangle 0 \end{aligned}$$

yields

$$\begin{split} \min\left(p_{acc} * \frac{C_{acc}}{C_r}, \ p_{acc} * p_r * \frac{C_{acc}}{C_r} + 1\right) \\ - \left[p_{acc} * \frac{C_{acc}}{C_r} + (1 - p_{acc}) * FPR + p_{acc} * TPR * \left(1 - (1 - p_r) * \frac{C_{acc}}{C_r}\right)\right] \right\rangle 0 \end{split}$$

 $\xrightarrow{\text{yields}} c > a * EPR + b * TPR$

(19)

where
$$a = 1 - p_{acc}$$
, $b = p_{acc} * (1 - (1 - p_r) * \frac{v_{acc}}{C_r})$, $c = \min(p_{acc} * \frac{v_{acc}}{C_r})$,
 $p_{acc} * p_r * \frac{c_{acc}}{C_r} + 1) - p_{acc} * \frac{c_{acc}}{C_r}$

The constraint function is linear, and the gradient of the constraint function is a constant. With additional constraints that $0 \le FPR \le 1$ and $0 \le TPR \le 1$, the feasible regions which make VoP >0 can be established. The feasible regions of FPR and TPR are different dependent on relative costs between being precautionary and ignoring risk. The shaded area with a distinguishable pattern is the corresponding feasible region for each, as shown in Fig. 6.

3.3. Risk thresholds

For a decision maker, a common situation is to determine which threats require a response and which threats can be tolerated. A threat should be responded if there is reasonably practical measure to reduce the risk even lower. Usually, we look at the potential consequence of the threat (accident cost), probability of the accidental scenario, or the combination, i.e., risk, to determine how to respond, as we do by using risk matrices.

Since the occurrence of accident is uncertain, being precautionary is a main principle to deal with the risk that we cannot afford to ignore [39]. The boundary condition between being precautionary and ignoring the risk can be derived from:

$$C(\text{be precautionary}) = C(\text{ignore the risk})$$
(20)

Meaning:

 $p_{acc} * p_r * C_{acc} + C_r = p_{acc} * C_{acc}$

According to Eq. (20), boundary values for accident consequence, probability, and risk between being precautionary and tolerate the risk can be obtained and they are:



Boundary

Fig. 7. Illustration of risk threshold between being precautionary and ignoring the risk.



Fig. 8. New boundaries when there is partial predictability.

$$C_{acc} = \frac{C_r}{(1 - p_r) * p_{acc}}$$
$$p_{acc} = \frac{C_r}{(1 - p_r) * C_{acc}}$$
$$p_{acc} * C_{acc} = \frac{C_r}{(1 - p_r)}$$

Decision maker should be precautionary and respond if p_{acc} , C_{acc} , or $p_{acc} * C_{acc}$ exceed the boundary value, which means being precautionary and respond is able to obtain a lower residual risk; otherwise, the risk can be tolerated and ignored as illustrated by Fig. 7. For example, consider the extreme case where the response action has 100% success probability, C_{acc}/C_r must be larger than $1/p_{acc}$ to make the response action rational and worth taken.

With prediction which provides positive value is involved, two new response boundaries are established, as illustrated by Fig. 8. One is upper boundary which is between being precautionary and trusting prediction; the other one is the lower boundary which is between trusting prediction and ignoring risk.

For the lower boundary, following the principle that C (with prediction) > C (ignore the risk), we get:

$$p_{acc} * C_{acc} + (1 - p_{acc}) * FPR * C_r + p_{acc} * TPR * (C_r - (1 - p_r) * C_{acc}) \rangle p_{acc} * C_{acc}$$

For scenario with potential accident consequence, the condition to tolerate the risk is:

$$\begin{split} C_{acc} &< C_r * \left[\frac{(1 - P_{acc}) * FPR}{(1 - p_r) * P_{acc} * TPR} + \frac{1}{(1 - p_r)} \right] \\ &= C_r * \left[\frac{(1 - P_{acc})}{(1 - p_r) * P_{acc} * LR} + \frac{1}{(1 - p_r)} \right] \end{split}$$

or

$$\begin{split} \frac{C_{acc}}{C_r} &< \frac{1}{(1-p_r)} * \left[\frac{(1-p_{acc}) * FPR}{p_{acc} * TPR} + 1 \right] \\ &= \frac{1}{p_{acc} * (1-p_r)} * \left[\frac{(1-p_{acc})}{LR+} + p_{acc} \right] \end{split}$$

When it comes accident probability for a known consequence, the condition to ignore the risk is:

$$p_{acc} < \frac{1}{1 + \left[(1 - p_r) * \frac{C_{acc}}{C_r} - 1 \right] * LR +}$$
$$= \frac{C_r}{(1 - p_r) * C_{acc}} * \frac{(1 - p_r) * C_{acc}}{C_r + \left[(1 - p_r) * C_{acc} - C_r \right] * LR +}$$

When it comes to risk (probability*consequence), the condition to ignore the risk is:

$$p_{acc} * C_{acc} < \frac{C_r}{1 - p_r} * \left[p_{acc} + \frac{(1 - p_{acc}) * FPR}{TPR} \right] = \frac{C_r}{1 - p_r} * \left[p_{acc} + \frac{1 - p_{acc}}{LR+} \right]$$

When prediction is involved, the minimum accident cost, minimum accident probability, and risk to initiate the response has changed. The availability of partial predictability pushes down the risk acceptance level. When prediction is involved and LR + > 1, the risk tolerance threshold is reduced for both probability and consequence. LR + > 1 is not difficult to achieve when VoP>0. The consequence tolerance threshold is reduced by a ratio of $\left[\frac{(1-p_{acc})}{LR+} + p_{acc}\right]$, and the probability tolerance threshold is reduced by a ratio of $\frac{(1-p_{r})*C_{acc}}{C_{r}+((1-p_{r})*C_{acc}-C_{r})+LR+}$; the risk tolerance threshold is reduced by $[p_{acc} + \frac{1-p_{acc}}{LR+}]$. The reduction ratio is in contrary with LR+, which means the better prediction, the high reduction of the risk tolerance threshold. Therefore, when there is no prediction, the risk tolerance threshold is quite high in order to initiate the response action; the consequence of the accident must be quite high compared with the action cost. It is not rational to initiate a costly response action for a threat with low risk, which meets the common sense.

For the upper boundary, following the principle that C (with prediction) >C (be precautionary), we get:

$$p_{acc} * \frac{C_{acc}}{C_r} + (1 - p_{acc}) * FPR + p_{acc} * TPR * \left(1 - (1 - p_r) * \frac{C_{acc}}{C_r}\right) \right\rangle p_{acc} * p_r$$
$$* \frac{C_{acc}}{C_r} + 1$$

For a threat with potential accident consequence, the condition to be precautionary is:

$$\begin{split} C_{acc} &> \frac{C_r}{1 - p_r} * \frac{1 - p_{acc} * TPR - (1 - p_{acc}) * FPR}{(1 - TPR) * p_{acc}} \\ &= \frac{C_r}{1 - p_r} * \frac{\text{Predicted negative (FN + TN)}}{\text{False negative(FN)}} \end{split}$$

When it comes accident probability with a known consequence, the condition to be precautionary is:

$$p_{acc} > \frac{1 - FPR}{\frac{C_{acc}}{C_r} (1 - TPR) * (1 - p_r) + TPR - FPR}$$

= $\frac{C_r}{(1 - p_r) * C_{acc}} * \frac{C_{acc} * (1 - FPR) * (1 - p_r)}{C_{acc} (1 - TPR) * (1 - p_r) + C_r (TPR - FPR)}$

When it comes to risk (probability*consequence), the condition to be precautionary is:

$$p_{acc} * C_{acc} > \frac{C_r}{(1-p_r)} * \left[\frac{1-p_{acc} * TPR - (1-p_{acc}) * FPR}{1-TPR} \right]$$
$$= \frac{C_r}{(1-p_r)} * \frac{\text{Predicted negative}(\text{FN} + \text{TN})}{\text{False negative}(\text{FN}) + \frac{1-p_{acc}}{p_{acc}}}$$

When prediction which provides positive value is involved, the thresholds of accident consequence, probability, and risk to be precautionary is pushed up. For accident consequence, the threshold increases by a rate of $\frac{1-p_{acc}*TPR-(1-p_{acc})*FPR}{(1-TPR)}$; the probability threshold is increased by a ratio of $\frac{C_{acc}(1-FPR)*(1-p_r)}{C_{acc}(1-FPR)*(1-p_r)}$; the threshold of risk level increases by a rate of $\left[\frac{1-p_{acc}*TPR-(1-p_{acc})*FPR}{1-TPR}\right]$. Therefore, it may not be necessary to be precautionary anymore if there is partial predictability that the decision maker can rely on for the same threat.

Risk thresholds established above only consider a single time point, namely the current time. The potentially improved predictability in the future is not considered. The established risk thresholds tell us "Should



Fig. 9. Decision tree diagram involving prediction for iceberg management.

we ignore the risk, or respond based on prediction, or just be precautionary and take action immediately?" without considering the option of postponing decision. In the same way, risk thresholds between responding now and responding later considering the whole-time spectrum can be established in a similar way.

3.4. Maximum response investment

Another situation that the decision maker needs to determine how much cost to invest in response action for accident prevention of a known threat. For a threat with a potential accident consequence of C_{acc} and occurrence probability of p_{acc} , what is the maximum cost that should be spent on accident prevention? In this case, we should compare the cost of taking action based on imperfect prediction (or the cost of being precautionary) with the cost of ignoring the risk.

When partial predictability is not considered, the condition to respond is:

C(be precautionary) < C(ignore the risk)

Meaning:

 $C_r < (1 - p_r) * p_{acc} * C_{acc}$

The maximum cost we should spend in accident prevention should be $(1 - p_r) * p_{acc}$ times of the damage that the accident could cause, which is much smaller than the potential accident consequence. Assuming $p_{acc} = 0.1$ and $p_r = 0$, the maximum cost should be invested for accident

Table 3

Costs of accident and response.

Prior collision probability (p _{acc})	Collision severity	Cost factor (C_{acc})	Action type	Cost factor (<i>C_r</i>)
	No collision Light	0 10	Disconnection Towing	50 1
0.1	collision Medium collision	300	-	-
	Severe collision	4000	-	-

prevention is only 10% of the accident consequence.

When the decision is made under partial predictability, the condition to respond is:

C(with prediction) < C(ignore the risk)

Meaning:

$$C_r < C_{acc} * (1 - p_r) * \left[\frac{p_{acc} * TPR}{p_{acc} * TPR + (1 - p_{acc}) * FPR} \right]$$

= (1 - p_r) * p_{acc} * C_{acc} * \left[\frac{LR +}{1 - p_{acc} + p_{acc} * LR +} \right]

The maximum cost we can spend in accident prevention is $C_{acc} * (1 - p_r) * p_{acc} * \left[\frac{LR+}{1-p_{acc}+p_{acc}*LR+}\right]$. It is $\left[\frac{LR+}{1-p_{acc}+p_{acc}*LR+}\right]$ times higher than the maximum responding cost when predictability is not considered. If the prediction is near perfect, the cost we can spend for accident prevention is slightly smaller than the damage of the accident. For accidents with a small probability, the investment in response action is much smaller than it would be if we do not consider the capability of prediction. The investment in response action increases after prediction capability is considered. Low response action investment based on risk (expected cost) is not reasonable anymore. Hence, predictability should be taken into account.

4. Case study on iceberg management

Iceberg management for collision avoidance is a time-critical task. If an iceberg is detected and approaching a Floating Production Storage and Offloading unit (FPSO), the offshore installation manager (OIM) needs to decide whether and when to respond. Iceberg collisions can be extreme events, depending on the iceberg mass and the drifting speed. Iceberg drifting is observable physical process that depends on weather and ocean dynamics. This process can only be partially understood and modelled. Therefore, whether collision will occur or not is only partially predictable. There have been many studies of iceberg trajectory prediction for iceberg management to prevent iceberg-offshore structure collision [3, 41]. To handle threatening icebergs and prevent collision, different measures can be taken dependent on the size of iceberg,



Fig. 10. Assumptions about prediction performance and towing failure probability in relation to time (prediction horizon).



Fig. 11. Expected cost of response actions with prediction horizons for light collision. (Note: for light collision, the costs of disconnecting action are much higher than the costs of towing action and ignoring the risk. Therefore, disconnecting is not shown in this figure.).



Fig. 12. Expected cost of response actions with prediction horizons for medium collision.

weather, and time available etc. [1, 42] It is common to monitor the iceberg and evaluate the risk of collision when the iceberg is far away [43, 44]. When the iceberg moves closer, and a collision is more likely to happen, an actual action is needed. For big and threatening icebergs, towing is the common response action. For small icebergs, water cannon and propeller washing are more commonly used. Disconnecting the FPSO is usually the last option, because of the high cost of disconnection [45]. Only towing and disconnection are considered in this case study.

4.1. Preliminary analysis

The decision-maker is to find the optimal response time and action to minimize cost. Fig. 9 illustrates the decision tree diagram for iceberg management. Being precautionary and ignoring the risk are included as options in the decision tree diagram.

4.1.1. List of assumptions and input data

The following assumptions in addition to those listed in Section 2.1 are needed in this case:



Fig. 13. Expected cost of response actions with prediction horizons for severe collision.

Table 4

Optimal prediction horison and action.

Collision severity	Optimal prediction horizon and action	Cost
Light	Towing the iceberg based on 8.5 hours' prediction	0.67
Medium	Being precautionary and towing as early as possible	2.4
Severe	Being precautionary and towing as early as possible	19.3

- The prediction performances are the same for the different types of icebergs.
- The prior collision probability p_{acc} is the same for all types of icebergs.
- Termination time t_t is unknown.
- Action costs are constant.
- The collision cost is dependent on the collision severity.

• The collision severity can be estimated with a satisfactory accuracy and is considered as known when an iceberg is detected far away and far earlier than the time to consider whether a response action is required. The collision severity is usually determined by the kinetic energy of the iceberg, which in turn is determined by the mass and velocity of the iceberg on the point of contact. However, the mass of the iceberg will not change much for 12 h because the deterioration rate is very small in cold region. The same applies to velocity. So, it is reasonable to assume the kinetic energy that the iceberg carries on the point of contact is the same as when it was observed. Assumed prior collision probability, costs for different collision severities and response actions are presented in Table 3. The assumptions are made based on the authors' judgment, therefore may not be accurate. For the preliminary analysis, relative costs (unitless) are used instead of the absolute costs of actions and accidents. This will not influence the results.

Assumptions about the prediction performance and action failure probability relative to time are shown in Fig. 10. The longest prediction horizon considered is 12 h. The minimum prediction horizon considered is 1 hour because the minimal time required for disconnecting is around 40–50 min [45]; and the minimal time required for towing is 4 h. Eqs. (21) and (22) represent TPR_{t_r} and FPR_{t_r} respectively, based on assumptions. Eq. (23) expresses the $p_{t_r,towing}$. The assumption about $p_{t_r,towing}$ is made based on judgement in combination with industrial data from Rudkin, Young [1] and Randell, Freeman [43]. Disconnecting is a costly but reliable response action and therefore has a lower failure probability than towing [45]. Eq. (24) expresses the assumed p_{t_r} disconnecting.

$$TPR_{t_r} = 0.3 + 0.6\log_{12}(13 - t_r), \ 1 \le t_r \le 12$$
 (21)

$$FPR_{t_r} = 0.2 - 0.15\log_{12}(13 - t_r), \ 1 \le t_r \le 12$$
 (22)

$$1 - \log_{10}(9), \text{ if } t_r \ge 12$$

$$p_{t_r, towing} = \{ 1 - \log_{10}(t_r - 3), \text{ if } 4 \le t_r < 12$$

$$1, \text{ if } t_r < 4$$
(23)



Fig. 14. Prediction effectiveness measured by the ratio between min (be precautionary, ignore the risk) and expected cost.



Fig. 15. Boundaries of TPR and FPR to enable positive VoP.

$$p_{t_{r, disconnecting}} = \begin{cases} 0.1 * (1 - \log_{13} 12), & \text{if } t_r > 12\\ 0.1 * (1 - \log_{13} t_r), & \text{if } 1 \le t_r < 12 \end{cases}$$
(24)

4.1.2. Results from the preliminary analysis

In this section, we describe the cost picture for 3 collision scenarios in a rank of severity/consequence: light, medium, severe collision. The cost pictures in relation to prediction horizon *t* for those 3 collision severities are shown in Fig. 11, Fig. 12 and Fig. 13 respectively. In each figure, the expected cost of imperfect prediction with two types of response action, the expected cost of ignoring risk, and the expected cost of being precautionary (lines with square markers) are presented. The expected cost of ignoring risk is shown by the dark blue line with double cross markers. The orange lines are for the towing action, while the blue lines are for the disconnecting action. By comparing the expected cost of ignoring risk, being precautionary and trusting imperfect prediction for the two actions, the optimal response time can be found, which is highlighted by a green dot in each figure.

If the assumptions match the real situation, the decision-maker can respond according to the calculated results by selecting the right severity level. Table 4 summarizes the optimal prediction horizon and action including its cost for each collision scenario. For light collision scenario, the best option is towing the iceberg based on 8.5 hours' prediction. For a medium collision scenario, the expected collision cost is 30, which is between towing cost (1) and disconnecting cost (50). The best option is being precautionary and towing the iceberg away as early as possible if there is more than 7.5 h available. The suboptimal option is disconnecting based on imperfect prediction with 1.5 h prediction horizon. For severe collision, the best action at the best time is towing the iceberg

anyway whenever detected if there is more than 10.5 h available. If there is less than 10.5 h available, the decision maker should be precautionary and initiate the disconnecting action. Postponing response and waiting for a better prediction is more costly than being precautionary in Fig. 13.

With the assumption that the response action failure probability is dependent on time available, and the prediction performance with different prediction horizon presented by Fig. 10, the decision cost of towing based on imperfect prediction matches the trade-off phenomenon, as illustrated by Fig. 2. Cost decreases with the increased prediction accuracy by using shorter prediction horizon in the beginning but increases if even shorter prediction horizon is used. This is because the increased failure probability of the towing action overweighs the benefit from improved predictability in the right part of the tradeoff curve. For disconnecting, the decision cost of disconnecting matches the left part of the trade-off curve because the failure probability of disconnecting does not increase as fast as the predictability.

It is better to be precautionary than to trust an imperfect prediction if the ratio of accidental cost and action cost is very high when considering the three collision scenarios. If the ratio between accident cost and action cost is small, it is better to ignore the risk than trust imperfect prediction. Applying and trust imperfect prediction is mainly useful for the case when accident cost and response cost are within a medium range, where the imperfect prediction to a large degree can provide added value (prediction effectiveness >1), as shown in Fig. 14.

The prediction performance requirement should vary for different accidental scenarios and response actions. Fig. 15 shows the boundary line of TPR and FPR which is calculated according to Eq. (16). For the



Fig. 16. Threshold of accident consequence and probability for towing action.



Fig. 17. Maximum response cost with/without considering predictability (the dot marks the maximum response investment).

 Table 5

 Optimal prediction horizon for optimal action with relation to collision cost.

Collision cost region	Response action (optimal time)
(0, 3.4)	Ignoring the risk
[3.4, 12.2)	Towing based on the optimal prediction horizon (8.5 h)
[12.2, 18.1)	Towing based on the optimal prediction horizon (9 h)
[18.1, 11,493)	Being precautionary and towing as early as possible when the iceberg is observed
$[11,493, +\infty)$	Being precautionary and disconnect as early as possible

case with accident cost 10 and disconnecting cost 50, there is no valid FPR & TPR which enable disconnecting response because disconnecting has a higher cost. The left region above each line is the feasible region of TPR and FPR for each specific severity scenario and response action (the ratio of accident cost to response action cost) with respect to 0 and 0.25 failure probability. For a high ratio between accident cost and response action cost, there is a high requirement towards TPR (true positive rate). This is because a failure to predict a severe accident will lead to a huge loss. On the other hand, for a low cost ratio, there is a high requirement towards FPR (false positive rate). Each false positive prediction (false alarm) will lead to big loss because of the wasted action. When the cost ratios are 6 and 10, FPR requirement become stricter when the failure



Fig. 18. Optimal response time of towing in relation to accident probability of light collision (collision cost is 10).



Fig. 19. Expected cost and towing action failure probability (the points with dot markers are the best responding time; the lines with markers are failure probability and the lines in the same color without markers are the corresponding expected cost).

probability increases. When the cost ratios are 80 and 300, TPR requirement becomes less strict when the response failure probability increases.

When prediction is not involved, taking action is the rational decision only if the cost ratio between accident and action is equivalent to or larger than 10 with assumed 10% chance of collision and 0% response failure probability. However, the cost ratio can be decreased when prediction is involved. With the assumed prediction performance, the accident cost threshold to disconnect the FPSO decreases to 84 (ratio =1.68) in the lowest (reduction factor of 4). For towing action, the accident cost threshold decreases to 3.4 in the lowest (reduction factor of 6.7). Fig. 16 presents the risk thresholds of towing action for ignoring risk and being precautionary when partial predictability is considered and when partial predictability is not considered. The thresholds of probability and accident consequence for ignoring risk is reduced while the thresholds for being precautionary is raised if partial predictability is considered.

At the same time, it is rational if more investment is required for the response actions with the assumed prediction performance, and response failure probability, as shown in Fig. 17. In Fig. 17, the maximum rational response investment is marked by dot. For light collision with a cost of 10, the maximum response investment is 0.95 which is lower than the towing action cost. Therefore, it is rational to ignore the risk when predictability is not considered. However, when partial predictability is considered, the maximum response investment increases to 3 which is higher than the towing cost. Then it is irresponsible to ignore the risk. The same occurs to disconnecting. For medium collision severity, it is better not to take disconnecting action when predictability is not considered since the maximum response investment is 30 and below the disconnecting cost 50. However, it worth to invest up to 180 for disconnecting when considering the predictability. The calculation provides rationale to judge whether the investment for accident prevention is reasonable or not.

4.2. Varying assumptions

Based on the results in 4.1.2, we can conclude that the solution provided by the VoP model is useful to find the optimal decision time point/prediction horizon for the planned response action. At the same time, the results show that the optimal decision time point is conditional on a number of factors, such as the ratio of accident cost to response cost, response failure probability, prediction performance, and accident probability. It is useful to know how the results change with different input data.

As the case study results, the optimal response time varies when it comes to different collision cost. Table 5 shows collision cost region and their optimal prediction horizons for the optimal action. It is rational to ignore the risk if when the collision cost is lower than 3.4 times of the

towing cost. Otherwise, if the collision cost is equivalent or higher than 18.1 times the towing cost, it is rational to be precautionary and take action as early as possible rather than use prediction. In the middle range, waiting for time with better prediction available gives better output. This means that the prediction will not always provide additional value and be useful. The higher the accident cost, the earlier response the action should be taken, and a more reliable response action should be chosen.

The prior accident probability also has an impact on the response time. Fig. 18 shows the optimal response time in relation to accident probability of light collision given towing as the response action. It is better to ignore the risk and not respond if the probability is below and equal to 0.025 (the optimal response time is 0). Further, it is better to be precautionary and respond as early as possible when the accident probability is above and equal to 0.173 (the optimal response time is 12). The prediction horizon should be longer which means that response decision should be made earlier if there is a higher accident probability when the accident probability is between the two boundary values. However, the optimal response time does not change dramatically and is anyway between 8 and 9 h under the current assumptions. The result here is only valid when the prediction performance is not impacted by the accident probability. Whether accident occur or not is skewed and not evenly distributed and this has an impact on the prediction performance; it is harder to obtain the same prediction performance for a small probability event; the FPR will increase to achieve the same TPR.

It is originally assumed that towing failure probability is close to the average performance of towing. However, whether towing can be successful is also dependent on ocean conditions such as wave condition, visibility and iceberg size. Fig. 19 shows the costs with five additional towing failure probability-time functions for the light collision severity scenario. New optimal response time is obtained which is marked by the large dot on each curve. The optimal response time is sensitive to the increase in action failure probability with decreased time available.

So far, there is no data available about prediction performance of any collision prediction model. There is considerable uncertainty about the future prediction performance. It would be necessary to see the impact of different prediction performance on the optimal response time and cost to obtain some indication of the uncertainty. Fig. 20 and Fig. 21 show the optimal response time and it's cost in relation to prediction performance. Basically, there will be a lower cost and at the same time it is better to respond earlier if the prediction is improved; TPR and FPR do not have the same degree of impact on the cost and response time.

In the preliminary analysis, only one response (one prediction & one action) is considered meaning no further action will be taken when the chosen action fails. In the case of an approaching iceberg, we may still be able to disconnect the FPSO if towing fails. Therefore, the expected collision cost may not be as high as only considering a single action available. The same applies to the false negative (wrongly predicting "no



Fig. 20. Optimal prediction horizons for varied prediction performances (1 is the originally assumed).



Fig. 21. Cost distribution and optimal prediction horizon in relation to prediction performance (the large dot markers indicate the optimal prediction horizon and 1 is the originally assumed prediction performance).

Table 6	
Optimal response strategy and the ex	xpected costs.

Collision severity	Optimal strategy	Optimal prediction horizons and the sequence of actions	Cost	Lowest cost with single response
Light	Single prediction	Towing based on 8.5 hours' prediction horizon.	0.67	-
Medium	Strategy 5	Towing first, disconnecting based on 3 hours' prediction horizon. Take a second prediction with 1 hour' prediction horizon if the first prediction predicts "no collision".	1.6	2.4
Severe	Strategy 5	Towing first, disconnecting based on 5.5 hours' prediction horizon. Take a second prediction with 1 hour' prediction horizon if the first prediction predicts "no collision".	2.7	19.3

accident"). It is interesting to see the result if a "series of responses" (multiple predictions and actions) strategy is taken. A second prediction can be implemented with a new optimal prediction horizon, assuming that whether the first implemented action is successful or not can be observed after it has been implemented for 2 h. Table 6 summarizes the new optimal responses when consider multiple prediction and actions.

Comparing with single prediction and action, taking a series of responses can make a large cost reduction for severe collision and the optimal response time changes also.

With the assumed data, for a light collision, making a single prediction with 8.5 hours' prediction horizon and towing based on the predicted results is the best option. For medium and severe collision scenario, it is optimal to take "strategy 5" described in Section 2.3, which is to initiate the towing action immediately and take prediction for the disconnecting and take a second prediction if the first prediction predicts "no accident" to compensate for errors in the first prediction. However, the prediction horizons for the disconnecting actions are not the same. For medium collision, a 3 hours' prediction horizon should be used and take the disconnecting action if the prediction predicts "collision". The expected cost of this strategy is 1.6 and less than the precautionary approach with an expected cost of 3.3. When a severe collision is expected, the best option is to use 5.5 hours' prediction horizon and to determine whether to take the disconnecting action. The expected cost of this strategy is 2.7 and less than the precautionary approach with an expected cost of 3.5.

When the collision cost is as high as 106, it is beneficial to consider a combination of two responses comparing with a single response with towing as the response action. If the collision cost is higher than 7078, being precautionary and take the series of actions as early as possible without any prediction is the best. The towing cost is 1/10 of the light collision cost, 1/300 of the medium collision cost, 1/4000 of the severe collision cost. The towing action is very cheap, so prediction is not



Fig. 22. Expected cost when the statues of towing action is observable and is not observable for the medium severity collision (Note: optimal response time is marked by the green dot.).

necessary to determine whether to tow for medium and severe collision. When a series of responses are planned, towing without prediction and only predict for the expensive disconnecting action and have a second prediction to compensate for wrong predictions. Prediction will be beneficial for the first towing action if the cost of the towing action is as high as 9.

Taking a series of responses can reduce the expected cost to a large degree when the accident consequence is severe, and the expected cost is much lower than a single response. For accident with severe consequences, expected cost of taking a series of actions without any prediction is much lower than the minimized cost when prediction is involved for a single action, which means that a low expected cost can be achieved by introducing a series of actions. Applying prediction and optimizing response time can reduce the cost even further. However, the expected cost will be higher as shown by Fig. 22 if the status (failure or success) of towing is not observable. Fig. 22 compares the cost pictures when the status of response action towing is observable and not observable for medium collision severity scenario. Comparison shows the importance of information related to the status of response action and cost reduction from the status information.

5. Implications and discussions

5.1. Implications for iceberg management

Considering an iceberg which is detected early and will cause a light collision, it is optimal to postpone the response and determine whether to tow the iceberg or not based on prediction made under 8–9 h prediction horizon. For a more severe collision, a series of responses would keep a low residual risk; towing should be initiated early. If the towing action fails, the decision whether to disconnect should be made based on 3–5 h prediction. To mitigate errors in the previous prediction, the last hour prediction can be made to determine whether disconnecting should be conducted if the previous prediction predicts "no collision". If the observed iceberg is very large, the consequence of collision is extreme and disconnecting the FPSO immediately is the best option.

Overall, prediction with good performance can reduce uncertainty and save unnecessary response actions, which eventually reduces the overall cost. If high prediction performance can be achieved for a longer prediction horizon, response decision can be made earlier, and cost can also be reduced. Prediction models should therefore be further improved. The decision can be postponed if there are ways to make the response action effective faster and more reliable. When a series of responses is considered, the information about response action status is critical and can reduce cost.

The result in the preliminary analysis is strongly influenced by our input assumptions. In the analysis, we compared three scenarios of collision in a range of cost with the assumption that the action failure probability and action cost is independent of the accident severity (accident cost). This assumption may not be reasonable in all situations. For larger icebergs, the towing action failure probability with the same amount of time available may be increased so that the response decision should be made earlier. A new failure probability function conditional on iceberg size and weather condition etc. can be used to calculate a more realistic response time.

One assumption is that iceberg collision can happen in any future unknown time point with a probability. There is also a deterministic model made to estimate time to collide or time to closest point of approach (CPA). Therefore, the time length of interest can be reduced if the deterministic model is used in combination with the prediction model. A waiting period that does not require monitoring can be calculated.

5.2. Implications for generic scenarios

The method in this paper answers the question when a decision should be made, and it can be used further for automated decisionmaking such as in autonomous systems. It is not always beneficial to postpone response and wait for more information except when the cost ratio between accident and action is in a moderate range. Therefore, decision making is actually easy when it comes to severe consequences or small losses. Following the basic principles of "being precautionary" or "ignoring risk" is good enough. Optimization of when to respond is not necessary. The middle part is where the decision maker tends to hesitate and may miss the optimal chance so that calculating the optimal response time is helpful. Risk boundaries between respond now or later can be established accordingly.

The prediction performance of different prediction horizons has an impact on the optimal response time. However, this impact is limited and dependent on how sensitive the prediction performance is over prediction horizon and how fast the response reliability decreases with time available. The response strategies will be in contradiction with each other for the two extreme cases 1) prediction performance does not increase for shorter prediction while failure probability increases fast, 2) prediction performance increases quickly while failure probability does not increase. For the first case, it is better to respond as early as possible if the risk from the threat exceeds the tolerance threshold. For the second case, it is better to postpone and respond late as long as there is enough time available for the response action.

Partial predictability should be considered. When partial predictability is involved and provide a positive value, risk is reduced, the minimum accident cost (risk tolerance threshold) for taking the response action is also reduced, the maximum response investment is raised. During operation, it is reasonable and possible that some predictability about major accident can be obtained than the design phase. Therefore, it is rational if more resources are required for accident prevention during operation. The risk tolerance threshold should be lower during operation than the threshold used in the design phase for the same threat. As claimed by the ALARP principle, risk should be reduced to a level as low as reasonably practicable. At the same time, the threshold to be precautionary is raised. There are both pros and cons for the increased threshold for being precautionary. The pros are that unnecessary action is avoided, and the saved resources can be used to create other values. The con is the lower down safeguards because reduced expected cost is not the same as increased safety. A careful verification and justification of the predictability is necessary.

In addition, the cost function which models the trade-off can be extended to optimize other parameters instead of time, such as:

- Distance, to calculate the optimal safe distance to take action (such as brake, give way) or not.
- Degradation degree, to calculate the optimal degradation degree to conduct preventive repairs or replace components.
- Information input (accuracy-processing load tradeoff, accuracynoise tradeoff), to calculate the optimal information input to form a judgement and initiate an action.
- Cutoff probability for risk monitoring. Many dynamic risk analyses are aimed at monitoring the accident probability and taking action when the probability increases to an unacceptable level. A cutoff probability needs to be defined to guarantee the best output (lowest cost).

The VoP model can be used to elicit prediction accuracy requirements for model development. In model development, it is a challenge to tune and select the optimal model to use. We foresee that the developed VoP model can be used for model development and a measurement of prediction performance (weighted performance indicator). The best model is the one that provides highest value of information instead of the highest accuracy. Because the costs of false positive and false negative predictions are different and "no accident"/ "accident" distribution is skewed, optimizing accuracy does not necessary obtain the least cost.

When a prediction model is used in a monitoring matter, a question we can ask is what the prediction frequency should be? In general, low prediction frequency (long monitor interval) can be used when an accident progresses slowly. In the case that the accident progresses faster, there should be a higher prediction frequency (shorter monitor interval). The VoP model perhaps can be used to optimize prediction frequency for accident monitoring also. If the resources required to obtain a prediction are high, this may be beneficial.

5.3. Information dynamics and use of information for accident prevention

New information does not necessary arrive by waiting. In addition, information does not necessarily enhance prediction performance. An important factor is how much predictability improves by waiting. Therefore, different possibilities and consequences should be included to determine whether waiting is beneficial in average. The method proposed calculates a statistical or Bayesian optimal response time which balances different possibilities.

In this paper, we discussed the increased prediction accuracy by waiting. Information about potential accident severity, weather conditions, barrier status etc. which influence response efficiency, can be collected also. Therefore, we may also include the prediction model and its error matrixes of a series of prediction horizons for potential accident severity and response efficiency etc. in the optimization model to obtain a refined optimal response time and action.

The value of information implicitly means the value of new information, but old information also plays a role in determining the value of new information. This implies that the time when information becomes available matters. In a real situation, we actively search for information and passively receive information from the environment. The understanding of importance of information dynamics may contribute to the design of an artificial information environment.

5.4. Limitations

This article excludes the situations where a threat is observed too late, or the accident occurs simultaneously with detection. In those cases, optimizing the time to respond is irrelevant because it is already too late. For the relevant and applicable situations, the challenges to the proposed method come from 1) input availability, 2) input accuracy and 3) assumptions. To calculate the optimal prediction horizon, time to decide, input data about the cost of accident, cost of action, action failure probability and prediction performance over time is required. Those inputs may not be easy to obtain. For example, we need information about the accidental scenario to know what action is suitable and resources we have to determine what action is feasible. The predictability-time function estimation needs a set of prediction error matrixes across all prediction horizons. For application, the function can be updated with new data from expert judgment, and/or experiments. Often, the cost of accident is difficult to estimate. However, the result will not change when the accident cost is extremely large (being precautionary is the best) or small (ignore the risk is the best). Accuracy in cost estimation is only required in the middle range.

In the article, we assumed that the prediction cost is negligible compared with accident cost and action cost. This may not be true in some cases, such as when it is expensive to obtain the inputs or expensive to construct a useful model. A prediction cost can be added to the calculation. In addition, accident cost and response costs are assumed to be known and constant. The model can be extended further considering costs as variants instead of as invariants. Varied prediction performance for each accident severity level can be included.

6. Conclusion

The observations made in this paper have wide application. For any risk control, response action must be taken ahead before accident occurs. When to respond, is a problem which needs to be answered for time-critical risk-control tasks. In this article, a Value of Prediction (VoP) model based on information value theory is proposed to calculate the optimal response time considering the trade-off between prediction performance and action failure probability over time. The optimal response time is dependent on the ratio between the accident cost and response action cost, accident probability, action failure probability, prediction performance and response strategy. An optimization of response strategy and time can maintain a low risk and high efficiency level. The results show that prediction does not always provide added value in accident prevention. When the accident consequence is extremely high, it is better to be precautionary and act as early as possible when the lowest failure probability is guaranteed, rather than trusting the prediction. When the consequence is comparable with the

cost of response action, then it is rational to ignore the risk. This confirms common sense. Calculating optimal response time based on VoP is supportive for threat which pose a moderate risk. Other important implications of imperfect prediction are that it can push down the threshold of risk acceptance and raise up the threshold of being precautionary and maximum response investment. Therefore, ignoring partial predictability is not proper. The VoP model can also be used to derive the prediction performance requirement for prediction model development.

CRediT authorship contribution statement

Tiantian Zhu: Conceptualization, Funding acquisition, Methodology, Writing – original draft, Writing – review & editing. **Stein Haugen:** Conceptualization, Funding acquisition, Project administration, Supervision, Writing – review & editing. **Yiliu Liu:** Funding acquisition, Supervision, Writing – review & editing. **Xue Yang:** Supervision, Writing – review & editing.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data Availability

No data was used for the research described in the article.

Acknowledgement

The first author would like to thank the China Scholarship Council for providing the funding, and Norwegian University of Science and Technology for the academic environment. The contribution from coauthor Xue Yang was supported by the National Science Foundation of China (NSFC) (Grant No. 52201408) and the Foundational Research Funds for the Central Universities (Grant No. 3132022154).

References

- [1] Rudkin P, Young C, Barron Jr P, Timco G. Analysis and results of 30 years of iceberg management. In: Proceedings of the 18th International Conference on Port and Ocean Engineering under Arctic Conditions; 2005. https://nrc-publications. canada.ca/eng/view/object/?id=7deaaa28-c001-4bf1-8261-1628c0802627.
- [2] Breitung J, Knuppel M. How far can we forecast? Statistical tests of the predictive content. J Appl Econ 2021:24. https://doi.org/10.1002/jae.2817.
- [3] Andersson LE, Scibilia F, Copland L, Imsland L. Comparison of statistical iceberg forecast models. Cold Reg Sci Technol 2018;155:69–89. https://doi.org/10.1016/j. coldregions.2018.07.003.
- [4] Zhu T, Haugen S, Liu Y, Hyungju K. Case study of major accident to demonstrate the possibility of prediction of conditions for accidents. PSAM14: Probabilistic Safety Assessment and Management. Los Angeles 2018. http://www.iapsam.org/p sam14/proceedings/paper/paper_48_1.pdf.
- [5] Gong Y, Liu P, Zhang J, Liu D, Zhang X, Zhang X. Considering different streamflow forecast horizons in the quantitative flood risk analysis for a multi-reservoir system. Reliab Eng Syst Saf 2020;204:107128. https://doi.org/10.1016/j. ress.2020.107128.
- [6] Dong, Y., J. Vinnem, and I. Utne. Towards an online risk model for dynamic positioning operations. Safety and Reliability – Safe Societies in a Changing World: Proceedings of ESREL 2018. ESREL. Trondheim, Norway. 2018, CRC Press. 10.1201/9781351174664-342.
- [7] Hogenboom S, Parhizkar T, Vinnem JE. Temporal decision-making factors in risk analyses of dynamic positioning operations. Reliab Eng Syst Saf 2021;207:107347. https://doi.org/10.1016/j.ress.2020.107347.
- [8] Dozza M, Boda C-N, Jaber L, Thalya P, Lubbe N. How do drivers negotiate intersections with pedestrians? The importance of pedestrian time-to-arrival and visibility. Accid Anal Prevent 2020;141:105524. https://doi.org/10.1016/j. aap.2020.105524.
- [9] Wang J, Yang F, Chen T, Shah SL. An Overview of Industrial Alarm Systems: Main Causes for Alarm Overloading, Research Status, and Open Problems. IEEE Trans Autom Sci Eng 2016;13(2):1045–61. https://doi.org/10.1109/ TASE.2015.2464234.

- [10] Dougherty EM. Time considerations in recovery analysis. Reliab Eng Syst Saf 1992; 35(2):107–15. https://doi.org/10.1016/0951-8320(92)90030-0.
- [11] Hill TP. Knowing When to Stop. Am Sci 2009. https://doi.org/10.1511/ 2009.77.126.
- [12] Cremen G, Bozzoni F, Pistorio S, Galasso C. Developing a risk-informed decisionsupport system for earthquake early warning at a critical seaport. Reliab Eng Syst Saf 2022;218:108035. https://doi.org/10.1016/j.ress.2021.108035.
- [13] Straub D, Faber MH. Risk based inspection planning for structural systems. Struct Saf 2005;27(4):335–55. https://doi.org/10.1016/j.strusafe.2005.04.001.
- [14] Zou G, Faber MH, González A, Banisoleiman K. Computing the value of information from periodic testing in holistic decision making under uncertainty. Reliab Eng Syst Saf 2021;206:107242. https://doi.org/10.1016/j. ress.2020.107242.
- [15] Chadha M, Ramancha MK, Vega MA, Conte JP, Todd MD. The modeling of risk perception in the use of structural health monitoring information for optimal maintenance decisions. Reliab Eng Syst Saf 2023;229:108845. https://doi.org/ 10.1016/j.ress.2022.108845.
- [16] Hollnagel E. Decisions about "What" and Decisions about "How". In: Cook M, Noyes J, Masakowski Y, editors. Decision making in complex environments. London: CRC Press; 2007. p. 10.
- [17] Zio E. The future of risk assessment. Reliab Eng Syst Saf 2018;177:176–90. https:// doi.org/10.1016/j.ress.2018.04.020.
- [18] Rothenberg DH. Alarm management for process control: a best-practice guide for design, implementation, and use of industrial alarm systems. Momentum Press; 2009. https://books.google.no/books?id=nPgqOMu0S4AC.
- [19] Yang X, Haugen S, Paltrinieri N. Clarifying the concept of operational risk assessment in the oil and gas industry. Saf Sci 2018;108:259–68. https://doi.org/ 10.1016/j.ssci.2017.12.019.
- [20] Wickelgren WA. Speed-accuracy tradeoff and information processing dynamics. Acta Psychol (Amst) 1977;41(1):67–85. https://doi.org/10.1016/0001-6918(77) 90012-9.
- [21] Heitz RP. The speed-accuracy tradeoff: history, physiology, methodology, and behavior. Front Neurosci 2014;8(150). https://doi.org/10.3389/ fnins.2014.00150.
- [22] Edwards W. Optimal strategies for seeking information: Models for statistics, choice reaction times, and human information processing. J Math Psychol 1965;2 (2):312–29. https://doi.org/10.1016/0022-2496(65)90007-6.
- [23] Fudenberg D, Strack P, Strzalecki T. Speed, Accuracy, and the Optimal Timing of Choices. Am Econ Rev 2018;108(12):3651–84. https://doi.org/10.1257/ aer.20150742.
- [24] Swets JA, Tanner Jr WP, Birdsall TG. Decision processes in perception. Psychol Rev 1961;68(5):301–40. https://doi.org/10.1037/h0040547.
- [25] Drugowitsch J, DeAngelis GC, Klier EM, Angelaki DE, Pouget A. Optimal multisensory decision-making in a reaction-time task. eLife 2014;3:e03005. https://doi.org/10.7554/eLife.03005.
- [26] Drugowitsch J, Moreno-Bote R, Churchland AK, Shadlen MN, Pouget A. The Cost of Accumulating Evidence in Perceptual Decision Making. J Neurosci 2012;32(11): 3612. https://doi.org/10.1523/JNEUROSCI.4010-11.2012.
- [27] Smith EE. Choice reaction time: An analysis of the major theoretical positions. Psychol Bull 1968;69(2):77–110. https://doi.org/10.1037/h0020189.
- [28] Howard RA. Information Value Theory. IEEE Trans Syst Sci Cybernet 1966;2(1): 22–6. https://doi.org/10.1109/TSSC.1966.300074.
- [29] Borgonovo E, Cillo A. Deciding with Thresholds: Importance Measures and Value of Information. Risk Anal 2017;37(10):1828–48. https://doi.org/10.1111/ risa.12732.
- [30] Bratvold RB, Bickel JE, Lohne HP. Value of Information in the Oil and Gas Industry: Past, Present, and Future. SPE Reservoir Eval Eng 2009;12(04):630–8. https://doi. org/10.2118/110378-PA.
- [31] Pratt JW. Introduction to statistical decision theory. Cambridge, Mass: MIT Press; 1995. ed. H. Raiffa and R. Schlaifer.
- [32] Yuan X-X, Higo E, Pandey MD. Estimation of the value of an inspection and maintenance program: A Bayesian gamma process model. Reliab Eng Syst Saf 2021;216:107912. https://doi.org/10.1016/j.ress.2021.107912.
- [33] Straub D, Ehre M, Papaioannou I. Decision-theoretic reliability sensitivity. Reliab Eng Syst Saf 2022;221:108215. https://doi.org/10.1016/j.ress.2021.108215.
- [34] Song C, Zhang C, Shafieezadeh A, Xiao R. Value of information analysis in nonstationary stochastic decision environments: A reliability-assisted POMDP approach. Reliab Eng Syst Saf 2022;217:108034. https://doi.org/10.1016/j. ress.2021.108034.
- [35] Bjørnsen K, Selvik JT, Aven T. A semi-quantitative assessment process for improved use of the expected value of information measure in safety management. Reliab Eng Syst Saf 2019;188:494–502. https://doi.org/10.1016/j.ress.2019.03.035.
- [36] Fawcett T. An introduction to ROC analysis. Pattern Recognit Lett 2006;27(8): 861–74. https://doi.org/10.1016/j.patrec.2005.10.010.
- [37] Arnborg S, Brynielsson J, Artman H, Wallenius K. Information awareness in command and control: precision, quality, utility. In: Proceedings of the Third International Conference on Information Fusion; 2000. https://doi.org/10.1109/ IFIC.2000.859871.
- [38] Sandin P, Peterson M, Hansson SO, Rudén C, Juthe A. Five charges against the precautionary principle. J Risk Res 2002;5(4):287–99. https://doi.org/10.1080/ 13669870110073729.
- [39] Sandin P. Dimensions of the Precautionary Principle. Hum Ecol Risk Assess Int J 1999;5(5):889–907. https://doi.org/10.1080/10807039991289185.
- [40] van der Helm HJ, Hische EA. Application of Bayes's theorem to results of quantitative clinical chemical determinations. Clin Chem 1979;25(6):985–8. https://doi.org/10.1093/clinchem/25.6.985.

T. Zhu et al.

- [41] Garrett C. Statistical prediction of iceberg trajectories. Cold Reg Sci Technol 1985; 11(3):255–66. https://doi.org/10.1016/0165-232X(85)90050-3.
- [42] Eik K. Review of Experiences within Ice and Iceberg Management. J Navig 2008;61 (4):557–72. https://doi.org/10.1017/S0373463308004839.
 [43] Randell C, Freeman R, Power D, Stuckey P. Technological Advances to Assess,
- [43] Randell C, Freeman R, Power D, Stuckey P. Technological Advances to Assess, Manage and Reduce Ice Risk in Northern Developments. Offshore Tech Confer 2009. https://doi.org/10.4043/20264-ms.
- [44] Edmond C, Liferov P, Metge M. Ice and Iceberg Management Plans for Shtokman Field. OTC Arctic Tech Confer 2011. https://doi.org/10.4043/22103-ms.
 [45] Fuglem M, Stuckey P, Huang Y, Barrett J, Thijssen J, King T, Ralph F. Iceberg
- [45] Fuglem M, Stuckey P, Huang Y, Barrett J, Thijssen J, King T, Ralph F. Iceberg disconnect criteria for floating production systems. Safe Extrem Environ 2020;2(1): 15–36. https://doi.org/10.1007/s42797-019-00007-4.